

ILLICIT FINANCIAL FLOWS AND TRADE MISPRICING: DECOMPOSING THE TRADE REPORTING GAP

p)

Alex Cobham Petr Janský Jan Mareš

IES Working Paper 26/2021



Opletalova 26 CZ-110 00, Prague E-mail : ies@fsv.cuni.cz <u>http://ies.fsv.cuni.cz</u>

Institut ekonomických studií Fakulta sociálních věd Univerzita Karlova v Praze

> Opletalova 26 110 00 Praha 1

E-mail : ies@fsv.cuni.cz http://ies.fsv.cuni.cz

Disclaimer: The IES Working Papers is an online paper series for works by the faculty and students of the Institute of Economic Studies, Faculty of Social Sciences, Charles University in Prague, Czech Republic. The papers are peer reviewed. The views expressed in documents served by this site do not reflect the views of the IES or any other Charles University Department. They are the sole property of the respective authors. Additional info at: ies@fsv.cuni.cz

Copyright Notice: Although all documents published by the IES are provided without charge, they are licensed for personal, academic or educational use. All rights are reserved by the authors.

Citations: All references to documents served by this site must be appropriately cited.

Bibliographic information:

Cobham A., Janský P., Mareš J. (2021): "Illicit Financial Flows and Trade Mispricing: Decomposing the Trade Reporting Gap" IES Working Papers 26/2021. IES FSV. Charles University.

This paper can be downloaded at: <u>http://ies.fsv.cuni.cz</u>

Illicit Financial Flows and Trade Mispricing: Decomposing the Trade Reporting Gap

Alex Cobham^a Petr Janský^{b,c} Jan Mareš^b

^aTax Justice Network ^bInstitute of Economic Studies, Faculty of Social Sciences, Charles University, Prague, Czech Republic ^cEmail (corresponding author): petr.jansky@fsv.cuni.cz

July 2021

Abstract:

Despite a Sustainable Development Goals target to reduce trade mispricing and other illicit financial flows, it is not clear how to measure trade mispricing over time for countries worldwide. We aim to combine a broad coverage of countries by using UN Comtrade data and robustness by developing a new methodology that sheds new light on a potential scale of trade mispricing for many countries worldwide. Specifically, we provide new estimates of the trade reporting gap and, for the first time, we decompose it into seven individual components. Our explorative analysis reveals three main findings. We show, first, that trade reporting gap is large, in absolute values as well as relative to the overall trade. Second, conceptually welldefined components such as product and country misclassifications account only for a small share of trade reporting gap. The large remaining residual hints at the degree of imprecision in international trade reporting and calls for a significant improvement in data quality. Third, the low-income countries' trade reporting gap has the highest ratio relative to their GDP, which is consistent with existing literature that shows low-income countries to be more vulnerable to a variety of illicit financial flows.

JEL: F13, F14, H26

Keywords: international trade; trade reporting gap; trade mispricing; illicit financial flows; low-income countries; global development

Acknowledgements: This research was supported by the European Union's Horizon 2020 program through the COFFERS project (No. 727145).

1 Introduction

When money flows out of countries illicitly, their economies shrink, government revenues fall and institutions are weakened (Brandt, 2020). Growing appreciation of the threat of illicit financial flows (IFFs) is reflected in the inclusion of a target to reduce them in the United Nations' Sustainable Development Goals (UNODC & UNCTAD, 2020), which might be one of the few ones that are likelier to be achieved after COVID-19 due to the government's increased desire for more revenues (Barbier & Burgess, 2020). Trade mispricing is one channel for IFFs and, as documented by numerous case studies, a real phenomenon. What is less clear is the scale of these IFFs and how important trade mispricing in reality is for low-income countries in particular.

In this paper we critically review the existing estimation methodologies that attempt to overcome the challenges posed by the inherently illicit nature of the studied phenomenon (Nesvetailova et al., 2021). One the one hand, some studies provide estimates for many low- and middle-income countries, but they are not very reliable, as is the case with the estimates by the Global Financial Integrity (e.g. Spanjers & Salomon, 2017). On the other hand, a few recent studies using confidential customs data have succeeded in confirming the existence of trade mispricing (Wier, 2020), but it is unrealistic to expect estimates for many low- and middle-income countries any time soon. Both of these streams of literature are unsatisfactory since they do not provide us with a reliable answer to the question of the scale of trade mispricing for countries at different levels of economic development. This poses an opportunity to develop a new methodological approach, which would combine the strengths of both – country coverage and reliability.

With this objective in mind, we carry out an explorative analysis using the UN Comtrade data. This guarantees us a good coverage of countries of all income levels and the challenge is to quantify trade mispricing in a reliable way. While we ultimately do not succeed in isolating trade mispricing from other phenomena, we discuss several interesting observations that represent systematic pricing patterns in international trade. Specifically, we consider a concept of trade reporting gap. We decompose it into separate and additive components such as product misclassification and trade costs, some of which

have been studied individually and some not, but, as far as we know, so far not studied together in this way. We follow up on our conceptual framework with empirical methodology that estimates, more or less roughly, each of the components that sum up into the overall trade reporting gap. For this trade reporting gap approach we highlight the limitations as well as opportunities for future research.

With this new empirical approach, we arrive at three main findings. First, trade reporting gap is large, in absolute values as well as relative to the overall trade. Second, conceptually well-defined components such as product and country misclassifications account only for a small share of trade reporting gap, whereas most of the trade reporting gap consists of the unexplained residual component. Third, the low-income countries' trade reporting gap has the highest ratio relative to their GDP and consists relatively more of unmatched trade, country misclassification and product misclassification components. This latter finding is consistent with existing literature that shows low-income countries to be more vulnerable to a variety of IFFs (Picciotto, 2013, Johannesen et al., 2020, Janský et al., 2021) and IFFs more generally hurting many countries worldwide (Hampton & Christensen, 2002) and likely benefiting only few (Butkiewicz & Gordon, 2013), highlighting the need for international policy reforms that work, in particular, for low-income countries.

Overall, our paper's contribution is threefold. We put trade mispricing in the framework of IFFs from low- and middle-income countries in particular. IFFs are included in the Sustainable Development Goals and, as with other goals, we expect them to likely shape - and be shaped by - the wider research community (Fontana & Oldekop, 2020). Second, we review and critically evaluate the existing methodologies to estimate trade mispricing. Last, but not least, we provide new insights into the trade reporting gap by carrying out an explorative analysis with the objective of decomposing the gap for as many countries as possible, with the main findings discussed above, and in more detail below.

The rest of the paper is structured in the following way. Section 2 provides a discussion of related literature in two parts – first, a more general review of IFFs from low- and middle-income countries and, second, a critical overview of existing methodologies to estimate trade mispricing. Section 3 introduces our preferred data, UN Comtrade, which has information on international trade from most,

or, arguably, all countries worldwide. Section 4 develops a new conceptual framework and estimation methodology to decompose the trade reporting gap. Section 5 provides an explorative analysis decomposing the gap into several components that shed new light on the potential scale of trade mispricing and related IFFs. Section 6 concludes, including with ideas for extending our empirical approach.

2 Related literature

In this section we introduce trade mispricing in the framework of IFFs from low- and middle-income countries and then we critically evaluate the existing methodologies to estimate trade mispricing.

2.1 Review of illicit financial flows from low- and middle-income countries

IFFs from low- and middle-income countries, and particularly from Africa (relative to its economic weight in the world), are estimated to be substantial. There are at least three reasons why IFFs are detrimental to development in low-income countries in Africa and elsewhere. First, IFFs lower tax revenues. Second, IFFs erode the funds available for private investment on the continent. Third (and probably most importantly), IFFs and associated activities, such as corruption (Tajaddini & Gholipour, 2018), pose a threat to the legitimacy of government institutions, which in turn leads to a reduction in tax morale, among other negative effects. Clearly, IFFs are inherently harmful and their effects are likely to differ by type and characteristics.

In recent years, IFFs and trade mispricing in particular have assumed greater importance on African policy makers' agendas. This is partly due to initiatives by non- and inter-governmental organisations such as Global Economic Governance (GEG) Africa, the Tana High Level Forum on Peace and Security in Africa (Cobham, 2014) and the African Tax Administration Forum (ATAF), which have stressed the development threat posed by trade mispricing in Africa. An important milestone in the policy arena was the release of an African Union Commission/United Nations Economic Commission for Africa (AUC/ECA) report or, in short, ECA (2015), which names IFFs as one of Africa's biggest development challenges. The Report of the High Level Panel on Illicit Financial Flows from Africa,

chaired by former South African president Thabo Mbeki, assesses the volumes and sources of illicit financial outflows, provides case studies on how these outflows occur in Africa, and recommends certain actions "that should be taken both by Africa and by the rest of the world to effectively confront what is in fact a global challenge" (ECA, 2015, p. 2). ECA is perhaps the leading policy actor in Africa when it comes to IFFs, evidenced in the number of reports released recently (ECA, 2018a, 2018b).

A useful classification of IFFs is distinguishing between four groups according to the underlying motivation (ECA, 2015). The first group involves market or regulatory abuse. The second group involves tax abuse. For these first two groups we expect private actors to feature most prominently, namely individuals, businesses and multinational enterprises (MNEs). The third group involves abuse of power, including the theft of state funds and assets. For this group politicians and public employees are likely to be the key actors. The fourth group involves crime and proceeds of crime, and here the most obvious actors are criminals. Within this typology, we expect that a large proportion of trade mispricing is associated with or results into tax abuse, including trade tariffs. While the typology may be a useful means of categorisation, there are as yet no corresponding estimates for the volume of IFFs per motivating factor, nor any data on the importance of trade mispricing compared with other IFFs.

The relative importance of trade mispricing and other types of IFFs still needs to be determined using reliable empirical methods. While clearly desirable to compare the different types of IFFs, attempts at quantification or decomposition have been limited. As far as we know, only rough estimates have been produced, such as those of Baker (2005) who is among the few who have attempted to estimate the scale of various types of IIFs at the same time. Drawing from a survey of managers and estimates by Baker (2005, p. 172), global annual lower-bound estimates of more than USD 1 trillion into commercial IFFs (USD 700 billion) and criminal IFFs (with drugs being the single largest criminal subgroup estimated at USD 120 billion) are decomposed. Within commercial IFFs, he distinguishes between trade mispricing (between unrelated trade partners, estimated at USD 300 billion) and fake transactions (no

trade actually taking place, estimated at USD 200 billion). According to these rough estimates trade transactions are responsible for most IFFs. Despite the anecdotal nature of the evidence, it highlights the potential prominence of trade mispricing in IFFs and supports the decision taken by Global Financial Integrity (GFI), an organisation that Baker founded, to base its estimates mostly on trade data. Although we discuss methodological limitations below, there are not many alternatives to estimating IFFs – and some of these, such as the UNECA methodology (ECA, 2015), build on the GFI approach. It is thus important to acknowledge the role of Baker and the GFI in shaping the public discussion on IFFs.

Trade mispricing is a well-known channel for IFFs. There are various definitions of trade mispricing (reviewed, for example, by Janský (2015). Trade mispricing (or mis-invoicing) is the deliberate overor under-invoicing of exports or imports by businesses in a country, which results into avoiding or evading tax or levies in that country. Transfer pricing is used by multinational corporations to price transactions between affiliates in different countries. The practice of transfer mispricing, also known as transfer pricing manipulation or abusive transfer pricing, is a subgroup of trade mispricing. It involves the manipulation of transfer prices – namely, interest payments, licence fees or payments for goods and services transferred between subsidiaries of the same multinational operating in different countries with regard to transactions that might be otherwise correctly recorded.

There is anecdotal evidence of trade mispricing by news outlets and governmental and nongovernmental organisations (NGOs), but systematic evidence of trade mispricing is scarce. Owing to data limitations and the illicit nature of IFFs, our knowledge of the extent and characteristics of IFFs is limited. The papers by the GFI and Ndikumana & Boyce (2001), who focus on African countries, and which are exemplified by the GFI's Spanjers & Salomon (2017) and Boyce & Ndikumana (2001), respectively, provide some of the most cited estimates of IFFs. The GFI methodology has some limitations but since it is widely used and there are no better estimates than those provided by the GFI, the authors of this paper have used them. Indeed, even the AUC/ECA (2015), which reports that Africa is estimated to be losing more than USD 50 billion annually in IFFs, argues that although these estimates fall short of reality, they draw attention to the scale of IFFs in Africa. Wier (2020), while focusing on one African country only (South Africa), persuasively documents that the trade mispricing is present and he estimated tax loss for the country at less than 1% of total corporate tax revenue.

More research is needed to formulate policies that will effectively and decisively reduce IFFs. Policy measures aimed at reducing IFFs need to be better informed by new and more focused research. This will entail acquiring high-quality trade data, ideally at transaction-level, with detailed characteristics of each transaction. Such data are not likely to be readily available in most African countries, with the exception, according to the literature, of Madagascar (Chalendard, Raballand, & Rakotoarisoa, 2017) and South Africa (Wier, 2020). Furthermore, the preparation of such data, which requires high-level technical skills and capacity, might prove challenging. We discuss these challenges in Section 5.

Governments and international organisations worldwide share a commitment to combat IFFs. In 2015 governments throughout the world agreed on the United Nations Sustainable Development Goals (SDGs). Led by low- and middle-income countries and African countries in particular, including the G77 and the African, Caribbean and Pacific (ACP) states, a consensus emerged that the elimination of IFFs should feature in the SDGs. As a result, one of the SDG targets (16.4) specifically addresses IFFs:

By 2030, significantly reduce illicit financial and arms flows, strengthen the recovery and return of stolen assets and combat all forms of organized crime.

Expert and political discussions on what indicators are suitable to follow up on this target are ongoing and it is not yet clear how progress will or should be measured. The selection of indicators could be crucial in obtaining more reliable data and a measure of IFFs. Cobham & Janský (2020) propose two indicators: first, how much profit multinational companies have shifted to achieve a tax-favourable misalignment with real activity; and second, offshore assets that are not declared to home country authorities. These two indicators focus on the inconsistencies resulting from IFFs rather than individual IFF channels, such as trade mispricing. Therefore, although the indicators do not explicitly capture trade mispricing, they can expose the results of trade mispricing. If the intention was to use an explicit trade mispricing indicator, possible options would be indicators similar to those refined by ECA (2015) or developed by frontier of research methods, including those by Wier (2020) for South Africa.

2.2 Methodologies to estimate trade mispricing

It is important to have reliable estimates of the nature and scale of trade mispricing for effective policy making. However, as discussed earlier, reliable estimates of trade mispricing are few and far between. In this section we critically review the most widely used methodologies for estimating trade mispricing, organised in four groups:

- 1. Pioneering methods;
- 2. Partner country trade statistics (or mirror trade statistics) method;
- 3. Price filter (or abnormal prices) method;
- 4. Frontier of research methods.

Table 1 summarises the main characteristics of each of these groups.

Table 1:	Four groups of	f methodologies u	ised to estimate ti	rade mispricing
----------	----------------	-------------------	---------------------	-----------------

Group	Prevailing	Prevailing	The gist of the	Recent	Robustness of	Availability	Coverage
	level of data	sources of	prevailing	examples	the	and country	of low-
		data	method		methodology	coverage	and
							middle-
							income
							income
							countries
Pioneering	Country	Non-trade	Various	Oxfam	Not robust	Large share	Covered
methods				(2000)		of the world	
	~			0.774	N		~ 1
Partner	Country	IMF	Comparing one	GFI's	Not robust, at	Most of the	Covered
country			country's	Spanjers &	the commodity	world	
trade			exports with its	Salomon	level, useful as		
statistics			partner	(2017)	indicative		
method			country's		information		
			imports				
Drice filter	Commodity	IIN	Identification	Chalandard	Not robust at	Most of the	Covered
	Commonly						Covered
method		Comtrade	of extremely	Raballand, &	the commodity	world	
			priced trades	Rakotoarisoa	level, useful as		
				(2017)	indicative		
					information		
Frontier of	Transaction	Country-	Systematic	Davies,	Potentially	Limited and	Limited
research		specific	differences	Martin,	robust, but	only a few	coverage
methods			between intra-	Parenti, &	only a few	countries	
			firm and arm's	Toubal (2017)	applications so		
			length prices		far		
			8 P5				

Source: Authors

Pioneering methods

The pioneering estimates (provided by a number of academics and NGOs, such as Baker, 2005) managed to put IFFs on research and policy agendas. For example, Oxfam (2000) argued that low- and middle-income countries suffered an annual tax revenue loss of USD 50 billion to tax havens, while Transparency International (2004) cited the billions of dollars in illicit financial flows that low- and middle-income country leaders were responsible for. Using the findings of Schneider (2005) and Murphy & Christensen (2005), Cobham (2005) estimated that low- and middle-income countries were losing USD 100 billion annually. Although pioneering estimates were important in attracting attention to topics closely related to IFFs in the past, they are of limited relevance today as they have largely been surpassed, in terms of credibility and coverage, by the methods discussed below.

Partner country trade statistics method

The partner country trade statistics method is likely the most frequently used method today. Its results have attracted the attention of policy makers to the reality and significance of IFFs, but its methodology is vulnerable to criticism. This method estimates the scale of IFFs by comparing the exports (imports) of a country (as reported by that country) with the corresponding import (export) figures supplied by the rest of the world in their trade dealings with the country. In this way it exploits the mirror nature of international trade data – that every flow is recorded twice.

The GFI regularly applies the partner country trade statistics method. The GFI focuses on low- and middle-income countries and provides estimates for African countries. Ndikumana & Boyce (2000, 2008) have frequently used the method to estimate capital flight from sub-Saharan African countries, while Nicolaou-Manias (2016) has applied the method to five African countries. A recent application of this method focused on trade-based money-laundering has been produced by Gara, Giammatteo & Tosti (2018) for Italy. We will concentrate on the GFI as its estimates have worldwide application and are likely to more credible. The most recent estimates by <u>GFI (2019)</u> for 148 countries for years 2006-2015 are based on the same methodology, the main change being the use of both Direction of Trade Statistics (DOTS) and Comtrade data to generate two sets of estimates.

The GFI results probably overestimate how much low- and middle-income countries lose in IFFs due to trade mispricing. This is mainly because of limitations in the partner country trade statistics method. It is not our goal to discuss the pitfalls of this method in detail, but a number of papers have criticised it, such as those of Nitsch (2012, 2016), Fuest & Riedel (2012), Hong & Pak (2016), Forstater (2016) and Nitsch (2017). Many of the method's limitations are linked to the availability of data. As it is difficult to obtain detailed trade data for many countries, the GFI relies on country-level data. As a consequence, it is necessary to make a number of assumptions which can make the estimates appear unreliable. Another limitation of the GFI's estimates is that they do not provide specific guidance on policy but constitute a general call to reduce IFFs.

The partner country trade statistics method has been refined over the years. ECA (2015) and other researchers have made improvements to this popular method, overcoming some of its earlier limitations. For example, instead of country-level data, ECA (2015) uses more detailed commodity-level data (although not as detailed or reliable as transaction-level data used by the frontier of research method discussed below). Whereas the method, as applied by the GFI, compares one country's trade with the rest of the world, Nicolaou-Manias (2016) uses bilateral data, which adds to the relevance thereof. With the GFI's application of the method it is assumed that there is no trade mispricing in partner (developed) countries. However, in reality, trade mispricing does occur to some extent, aimed at transferring funds from advanced economies as well (Hong & Pak, 2016). Indeed, Kellenberg & Levinson (2019) find evidence of trade misreporting for countries across all income groups.

Because of the dearth of detailed data, the GFI's Spanjers & Salomon (2017) assume that transportation costs are 10%, whereas in reality they differ across countries and even transactions. Nitsch (2016) criticises this assumption and Erskine (2018) finds that transaction costs are systematically higher for landlocked countries. To remedy this, ECA (2015) follows a country-specific approach.

While ECA (2015) and other researchers have improved the method and partially overcome certain problems of the partner country trade statistics method, other limitations persist. Of course, the

significance of the limitations depends on the research question to be answered. Where indicative or illustrative answers will suffice, the most improved versions of this method could prove useful, but for estimates of overall scale, the method remains unreliable.

Price filter method

The price filter method identifies extreme or abnormal prices in international trade, which signal IFFs. The method starts by estimating prices as unit values by dividing the financial values by kilograms. It then establishes what a normal price for a given commodity should be and labels any prices outside this filter as abnormal and the associated trade flows as illicit. This pivotal assumption implies that any abnormal prices are related to IFFs, which is likely to lead to overestimation. Consequently, we do not consider it a reliable method for estimating the scale of IFFs. Still, it can be useful for other, indicative purposes, such as identifying cases warranting a more detailed audit.

A number of academics have applied the method, notably Pak & Zdanowicz and their co-authors. These include De Boyrie, Pak, & Zdanowicz (2005), Pak (2007) and Zdanowicz (2009). In addition, NGOs such as Hogg et al. (2009, 2010) have used various versions of this method (including the example of Switzerland-Zambia trade in copper, an extractive industry for which <u>Ponsford & Mwiinga</u> (2019) document Zambia's government's request for financial models from companies). For example, De Boyrie, Pak, & Zdanowicz (2005) used transaction-level data for trade between the US and Russia and attributed the flows through trade mispricing to money laundering and tax evasion. They estimated that in 1995 the amount of capital shifted out of Russia in the form of abnormal prices was 3% of total exports and 6% of total imports, respectively.

A number of authors have criticised the price filter method, e.g. Carbonnier & Zweynert de Cadena (2015). We agree that the assumptions are too strong and the estimates are not particularly helpful in determining the overall scale of trade mispricing. However, we believe that the method is useful for highlighting the specific commodities and countries most vulnerable to trade mispricing and might become more reliable in the future if further developed and applied to the best available data.

In addition, the World Customs Organisation (2018) presented its study report on IFFs and trade-misinvoicing to the Development Working Group of the G20 in July 2018. The multi-co-authored report argues that estimates of both partner country trade statistics and price filter methods are not sufficiently robust and should not be understood as a reliable quantitative measurement of the scale of IFFs, but rather as a risk indicator, which can be useful in comparing the risk of IFFs across commodities, countries and over a longer time period. The World Customs Organisation (2018) also makes the important point that rather than disputing the accuracy of individual assessment mechanisms, attention should instead focus on actions to combat IFFs, the existence of which is indisputable, the estimates of which, however, are dependent on the methodologies used.

Frontier of research methods

Frontier of research methods have so far provided the most rigorous evidence of trade mispricing. These methods are rigorous in their approach and are applied to detailed transaction-level data. To date they have provided some of the most convincing evidence of trade mispricing, mainly in developed countries. Studies for the United States have been conducted by Clausing (2003), Bernard, Jensen, & Schott (2006) and Flaaen (2017); for France by Davies, Martin, Parenti, & Toubal (2017) and Vicard (2015); for Denmark by Cristea & Nguyen (2016); and for the United Kingdom by Liu, Schmidt-Eisenlohr, & Guo (2017). The only directly comparable study for a low- or middle-income country is that by Wier (2020) for South Africa, which provides the first direct, systematic evidence of profit shifting through transfer mispricing in such a country.

There is also a clear recommendation emerging from the frontier of research papers: the tax authority should set up an automated flagging system. This digital system would automatically test for deviations in the pricing of related and unrelated transactions. Building on his experience of South Africa, Wier (2020) argues that this is a cost-effective way of curbing transfer mispricing as it uses information that is abundant but not efficiently exploited. He proposes that when a firm prices a product differently in related and unrelated transactions, this should prompt an automatic audit or, as a minimum, a flag should be raised and an email sent to the firm cautioning them to stop this behaviour.

He argues that the cost of doing this is in the thousands of dollars whereas the potential tax gain is in the tens of millions of dollars.

Overall, the evidence of trade mispricing is of relevance to policy makers, but it could be much improved. While the not-so-reliable partner country trade statistics and price filter methods remain the only ones that have produced results for many low- and middle-income countries, we argue that attention should be given to the application of new methods, such as frontier of research methods discussed. However, given the data requirements of the frontier of research methods that are unlikely to be met in many low- and middle-income countries, it is still worth seeing what can be done with the data is widely available, UN Comtrade data. We develop one such approach that combines some aspects of the partner country trade statistics and price filter methods.

3 Data

We primarily rely on the UN Comtrade data, the most comprehensive database of international trade in goods. The database records annual bilateral trade on product-level by more than 150 countries. UN Statistical Division applies the Harmonized System (HS) product classification which at its most granular level (HS 6-digit) distinguishes about 5300 categories and contains roughly 15 million bilateral flows every year. In this paper, we employ the finest classification to explore trading and potential trade mispricing patterns. We mostly focus on 2015, the year with the best data availability at the time of the empirical analysis for majority of the reporting countries, but we also show the dynamics of trade gap components since 2010 capture the trade gap and its components over time. While we could potentially use even more recent years in UN Comtrade, the data is getting sparser as some of the countries lag on their reporting. We also follow the criterion of data availability in the choice of HS classification where we rely on the 2002 vintage. Not all the reporters offer data reports in the latest versions, which would limit our sample. Additionally, this vintage also ensures better backward compatibility of our estimates and enables extensions all to way to 2000s'. In the analysis, we combine the UN Comtrade data with the country classifications and other economic indicators from the World Bank. We discuss descriptive statistics of the UN Comtrade data set in the Appendix 2.

One important consideration working with UN Comtrade is the fact that there is a difference in the prices reported on imports and exports. The export prices are generally declared as Free on Board (FOB) while the import prices as including Costs, Insurance, and Freight (CIF). The CIF prices should typically be higher than FOB prices with the difference range from 10 to 20% according to the World Bank (2010). Accounting for such differences is important when constructing mirror trade data or attempting to fully utilize the trade flows data collected by the reporting countries. The optimal approach towards reconciliation of import and export prices is unclear. Although the number of reporting countries has been growing, there are a lot of jurisdictions which do not provide data on their trade flows. In practice, around 150 countries report the data, but there are more than 200 partner jurisdictions reported in the database.

This presents us with further choices when analysing the data. One of the possibilities is the estimation of additional CIF costs and adjusting to their FOB equivalents. Gaulier and Zignago (2010) introduce such approach which results into BACI database of international flows. The average CIF estimated in BACI is around 3% which is somewhat lower than typically assumed (and lower than the price differential suggested by the UN). Although BACI offers a consistent database on mirror flows, the adjustment of mirror trade prices during its construction might mask some of the potential mispricing in the original data. We therefore stick with the original UN Comtrade data and use existing (WB trade cost data, CEPII, and also our own assumption of 5%) estimates of these factors and a sensitivity analysis to estimate the scale of this phenomenon in a separate component in the decomposition analysis below.

Differences in declared export and import values when using mirror data may also arise from other sources. The commodities can be classified into non-corresponding categories at HS 6-digit level by import and export customs administrations. Other discrepancies may arise, for example, due to the trades taking place around the turn of the year. The severity of these distortions cannot be precisely

investigated, but below we acknowledge and empirically isolate as many of these possible effects. Below we introduce a definition of trade reporting gap and methodology we apply to the UN Comtrade data to decompose it.

4 A simple economic model of trade misreporting incentives

We present in this section an economic model of firm-level traders' misreporting behaviours that takes into account the characteristics of bilateral, product-specific trade flows to guide our empirical analysis in the following sections. This model is the generalised version by Kellenberg & Levinson (2019) of the model in Ferrantino, Liu, & Wang (2012), which in turn builds on modelling the transfer pricing problem within multinational firms such as Swenson (2001) and Bernard et al. (2006).

We assume that for each product *i* and year *t*, there is a representative exporting firm in country *x*, and a representative importing firm in country *m*. The two firms arrive at the true value of free-on-board (FOB) exports, V_{xmit}^* , which is unknown to customs officials. Each exporting and importing firm decide how much of its exports to report to the customs, V_{xmit}^x and V_{xmit}^m , respectively. We let δ_{xmit}^x and δ_{xmit}^m be the proportional deviations from the true values of exports and imports by exporters and importers:

$$V_{xmit}^x = \delta_{xmit}^x V_{xmit}^*$$
 and $V_{xmit}^m = \delta_{xmit}^m V_{xmit}^*$

Exporting firms may not report any exports at all $(\delta_{xmit}^x = 0)$, underreport $(0 < \delta_{xmit}^x < 1)$, report accurately $(\delta_{xmit}^x = 1)$, or overreport $(\delta_{xmit}^x > 1)$. Similarly, importing firms may not report any imports at all $(\delta_{xmit}^m = 0)$, underreport $(0 < \delta_{xmit}^m < 1)$, report accurately $(\delta_{xmit}^m = 1)$, or overreport $(\delta_{xmit}^m > 1)$. These deviations can be in the form of misreporting prices or quantities. There are a number of reasons why firms would misreport exports or imports and in the following we discuss how we distinguish between them and how we estimate their scale from the available data.

5 Conceptual framework and estimation methodology

In this section we develop a new conceptual framework that decomposes the trade reporting gap intro individual components and we introduce the methodology to estimate the scale of each of the components. The trade reporting gap (or misreported trade or reported trade gap or, simply, gap) has been studied in the literature, as reviewed above, most famously by the Global Financial Integrity and also recently by Kellenberg & Levinson (2019). However, none of the research has so far attempted such detailed analysis of the gap's various components as we do now and this requires to redefine the basic terms and develop a new conceptual framework that allows for the various components and the use of more detailed data than the previous research has used. We conceptually think of the gap and also empirically estimate it at the highest level of disaggregation possible (HS6 commodity groups in the UN Comtrade). In addition, we aggregate the results at higher levels (countries, regions, world) to be able to present the results and to provide further insight on the relative importance of individual misreporting categories as well as to relate the estimates to country characteristics.

We define the trade reporting gap as the overall absolute value of misreported trade between two trading partners. The fact we use the absolute value makes it more practicable for a decomposition, although it makes it less directly comparable with some of the existing literature. By construction, the seven components are additive and we sum up them in absolute values into an overall value of the trade reporting gap. In simple terms without detailed notation (for countries, years and products), the decomposition can be described as:

Trade reporting gap

- = |Country misclassification| + |Product misclassification|
- + |Unmatched trade| + |Abnormal prices| + |Trade costs| + |Tariff costs|
- + |Residual|

For each of the seven components, we briefly explain the rationale for its existence and we outline how we estimate the scale of the components with the UN Comtrade data in Table 2.

In principle, the gap between the reported trade values can arise due to one of the trade partners not reporting the trade flow at all (unmatched data, corresponding to components 1-3), or due to the

difference in the declared value by the trade partners (matched data, corresponding to components 4-

7).

Number	Component	Estimation strategy
1	Country	Assuming HS6/4 correct classification, unmatched exports from an exporter
	misclassification	that overlap with unmatched imports from an exporter are a proxy for this
	(different partner,	misclassification (and a method for identification of the three-way transit
	transit trade)	trade, not visible but derivable from the data). This is the only common
		component for exports and imports.
2	Product	Misreported trade scale larger at the HS6 than the HS4 levels (and the
	misclassification	difference is the scale of the misclassification difference). (Plus doing this
	(different HS6)	only at the specific country-pair level and inputting the transit trade on the
		basis of the reported trade shares, which is an assumption.)
3	Unmatched trade	HS6 exports not reported at all by importers.
	(no corresponding	
	at HS6, even after	
	controlling for	
	misclassifications)	
4	Abnormal prices	We estimate this as a residual for which there are extreme, abnormal prices
	(includes also	(with an inevitably arbitrary cut-off guided by statistical literature, if the
	deliberate over and	price is more than two standard deviations away from the world weighted
	under pricing)	average).
5	Trade costs (CIF,	We use an assumption of 10%, which is in line with existing research. For
	FOB, landlocked,	example, the CIF prices should typically be higher than FOB prices with the
	as obvious,	difference range from 10 to 20% according to the World Bank (2010).
	systematic reasons	
	for differences in	
	prices)	
6	Tariff costs	Data on tariffs from WITS TRAINS database. We apply Most-Favoured-
		Nation (MFN) tariff rates of importing country against the countries

 Table 2. Decomposing the trade reporting gap: the seven components

		globally on the import flows by HS6 category.
7	Residual	We estimate this as a residual, i.e. what cannot be explained by other
		components. This might include anything, including deliberate misreporting
		of amount or partially unmatched at transaction level (parts of the flow not
		reported by one partner). Any transit trade that we are unable to identify
		using the triangular trades.

Source: Authors

Country misclassification

An important reason why unmatched data may occur in the database is the transit trade. By convention, exporting countries declare the next country where the trade flow is directed. However, this might not be the terminal stop, but merely a trading hub, through which the goods flow (either physically or "on paper") to another country. Nevertheless, at the destination country, such trade flow gets reported with the origin jurisdiction as a trade partner. This way of recording the trades implicitly introduces mismatches into the data. We try to partially account for this transit trade by mapping the unmatched exports to unmatched imports and netting them off. Unmatched import is any trade flow that a country reports as an import, but the trade partner declared for this flow does not report it as an export. Similarly, unmatched export is an export, where the declared trade partner does not report the flow as an import.

We first identify all the exports going out of a country which do not have a corresponding reported import (same HS6 category and trade partner), since these flows might be unmatched due to transit trade. Then we identify unmatched imports to all the jurisdictions and sum them across importing jurisdictions for all partners (exporters) and HS6 commodity codes. This is the potential "end" side of the transit trade. We join the latter sums with the exporters' data on unmatched exports by commodity categories and lessen the unmatched export flows proportionally to the potential transit trade and the share of specific trade flow on the overall unmatched exports within individual HS6 commodity category. We then sum up absolute differences for a country pair with the adjusted trade value. We follow the equivalent procedure for the imports. Estimating the trade gap with and without accounting for transit trade serves as an estimate of the amount of trade misreporting arising due to existence of transit trade (component 1). Admittedly, this allowance for transit trade is not perfect as the unmatched flows do not net off perfectly and nonnegligible volumes of trade remain unmatched.

Product misclassification

Since we look at the most granular level of the database and HS6 commodity categories, it might be the case that the trade flow is reported in different 6-digit categories (this might be, for example, either as a consequence of an error, the two countries' customs officials disagreement, or deliberate attempt to evade tariffs by the trader's misclassification from high- to low-tariff category). We estimate the size of this reporting gap by computing the difference of our baseline unmatched trade volume with the volume of unmatched trade when using 4-digit level classification of the data. In theory, it can happen that the customs of two countries categorize a trade flow in non-corresponding HS6 categories, but it is much less likely that they will categorize the same trade flow in different HS4 categories which tend to be much broader. The unmatched trade at the higher level of aggregation must always be lower by definition and the difference between the estimates captures the volume of misreported trade due to misclassification at HS6 level (component 2).

Unmatched trade

The third component and category accounting for unmatched data are the *true* unmatched flows, which we label as unmatched trade (component 3) and which we estimate as the overall unmatched flows less the mismatches due to misclassifications in the data caused either by transit trade (component 1) or different HS6 classification (component 2).

Abnormal prices

We distinguish between four components of trade misreporting in the matched data and the first category which we can empirically capture is the misreporting of trade through extreme pricing (component 4). We use the matched trade flows to compute the average prices for commodities at HS6 level along with their standard deviations. We then compare the prices of individual flows with these

reference prices and classify the trade into extreme prices category if the price is more than two standard deviations away from the world average (a rule of thumb for statistical significance). This choice is arbitrary, and the definition of abnormal prices naturally affects our results. The sum of differences between trading volumes declared by corresponding trading partners is the misreported trade in this category. After accounting for extreme pricing, the remaining misreported trade falls in three categories. The value that captures the costs of trade which are only recorded by one side of the transaction (component 5), tariff costs (component 6) and the residual misreported trade we cannot explicitly classify (component 7).

Trade costs

The former, trade costs, arises not only due to the different recording standards (CIF vs. FOB), but also due to differences in transaction costs of international trade, which might include systematic factors related to country characteristics (countries that are landlocked or members of a free trade area are likely to have higher import prices relative to countries that have access to the sea and low tariffs with their trading partners) as well as goods characteristics (country importing goods with higher than average weight will have high transaction costs), but also some non-systematic, random factors. One way to account for trade costs and one that we use for headline results is to assume a fixed proportion of trade flow values to be trade costs.

To estimate the volume of the misreported trade of this nature we assume fixed costs of trade of 10% of the value declared by the exporter – FOB (technically, we net off the categories detected as abnormal in prices). The assumed 10% is on the lower bound of the World Bank (2010), but higher than Gaulier and Zignago (2010). The United Nations Economic Commission for Africa & African Union's (2015, page 95) ECA model uses the BACI database, which provides reconciled bilateral trade flows using Comtrade data at the HS6 level of product disaggregation, explained in detail by Gaulier and Zignago (2010). They contrast this their Trade Mispricing Model, which uses a fixed CIF/FOB ratio of 1.1 for assessing the value of CIF. The GFI's Spanjers & Salomon (2017) used to assume 10%, but the most report by <u>GFI (2019, page 3)</u> assumes 6% for the IMF DOTS data and in this

follows the IMF researchers working on DOTS data, who newly use it to produce CIF/FOB-adjusted values for non-reporting countries (Marini, Dippelsman, & Stanger, 2018, page 11). In turn, IMF researchers explain their assumption by using a cross-country average coming from OECD's Miao & Fortanier (2017), who prepare the bilateral, product level international trade and insurance costs for more than 180 countries and partners, over 1 000 individual products, for the 1995-2014 time period. Similarly, Duval, Saggu, & Utoktham (2016) estimate trade costs using a regression model.

Tariff costs

To estimate tariff costs, we use data on tariffs from a leading database WITS TRAINS, discussed, for example, by Anderson and van Wincoop (2004) or Shepherd (2010), and sourced from UNCTAD and the World Bank. We apply Most-Favoured-Nation (MFN) tariff rates of importing country against the countries globally on the import flows by HS6 category to estimate the scale of this component.

Residual

What we are left with is the residual misreported trade which contains the misalignment at the level of individual transactions, unintended reporting errors, and the transit trade we are not able to account for using the mapping between unmatched imports and exports.

6 **Results**

We decompose the overall trade gap to seven components described in the section on methodology. After identifying the trade gap components on bilateral basis, we aggregate the results on country level. For the ease of exposition, we also aggregate the country level data by income levels and geographic regions following the classification of the World Bank as this allows us to show patterns in trade gap components across otherwise heterogenous country observations. Table 3 presents the estimated decomposition of the misreported trade summed up across countries for 2015. With the important exception of the unexplained residual of 63%, we ascribe the highest share of the differentials to trade costs. Our estimates suggest that the trade costs might make up nearly 15% of misreported trade. To some extent, this share is driven by our choice of fixed costs. We only account

for the costs when analyzing matched trade flows, as we treat the we treat the unmatched trade as strictly misreported. Along with the higher share of trade gap resulting from matched trade flows than the unmatched in the overall misreported trade, this explains why the share of costs on the reported trade gap is higher than assumed fixed costs of 10%.

	Component	Estimated scale (billion USD)	Share (% of trade reporting gap)
-	Total trade reporting gap	9507	100
1	Country misclassification	168	1.8
2	Product misclassification	656	6.9
3	Unmatched trade	837	8.8
4	Abnormal prices	70	0.7
5	Trade costs	1409	14.8
6	Tariff costs	380	4
7	Residual	5987	63.0

Table 3. Decomposing the trade reporting gap, world totals, 2015

Source: Authors

The remaining notable components of the gap are unmatched trades, product misclassification and tariff costs. Unmatched trade represents 8.8% of the world trade gap. It is important to note that we only partially address the transit trade and some of it might still be hidden in this component (some of the transit trade may also appear in the commodity categories and country pairs we identify as matched and in these we do not make any adjustments). This number should thus serve as an upper bound for the "true" unmatched trade. Product misclassification constitutes 7% of the world trade gap. It is likely that this share would increase if we accounted for misclassification at higher HS levels of aggregation at the expense of unmatched trade. The tariff costs are estimated to account for 4% of the trade reporting gap. The remaining components, abnormally priced trade flows and country misclassification, together account for less than 3% of the trade gap. Overall, conceptually well-

defined components such as product and country misclassifications account only for a small share of trade reporting gap and most of the trade reporting gap consists of the unexplained residual component.

We show the extent to which estimates of trade reporting gap are downward biased at higher levels of aggregation. Figure 1 shows the comparison of overall trade gap when we compute at different levels of HS aggregation. The misreported trade decreases continuously all the way to HS0, where we compare the trade flows summed across all commodity categories for mirror country-pairs. The value of the trade gap shrinks down to 4089 bn. – less than half of the trade gap estimated at the HS6 level. While the trade reporting gap estimated at various levels of aggregation reveals that the gap substantially increases the more detailed the aggregation is, we consider the most detailed aggregation preferable and why we use it in our decomposition results. In any case, trade reporting gap is large, in absolute values as well as relative to the overall trade. For an illustrative comparison, we note that the total sum of all exports and imports in 2015 is 29707 billion USD, but this sum is double counting most of the flows, which appear as both exports and imports. Our indicators of trade reporting gap are designed to be largely without this kind of double counting and thus these two are not directly comparable. As a more comparable total trade value could be considered the sum of all exports or imports or half of the combined total.

Our estimates allow us to explore the importance of trade gap components for individual countries. Table 4 presents the list of top country standings in each of the decomposition categories. We compute the shares of individual components on the overall trade gap and then reorder the countries in each component. In addition, Table 5 then introduces a straightforward indicator of trading hubs based on dollar value of unmatched exports, i.e. only a part of the unmatched trade component. The values of unmatched exports are net off product misclassification and transit trade and capture only the unmatched trade for partner countries declared in identified unmatched exports. In other words, the countries in this table are often declared as partners in trade which is only reported by the exporter. Essentially, such countries might either be transit countries (which might be considered trade hubs), or deliberately not report trade for other reasons.

There are substantial differences across income groups: high-income countries account for much of the trade reporting gap in the absolute values, while tiny low-income countries' gap has the highest ratio relative to their GDP. Figure 2 presents trade reporting gap and its components as shares overall trade gap (%) for each of the income groups for 2015 (other figures, with estimates over time and across regions as well as across income groups in absolute values and relatively to GDP, are included in the Appendix 1). Interestingly, the low-income countries' gap consists relatively more of country misclassification, product misclassification and unmatched trade (although for this latter one, the upper-middle income countries have a larger share) components.

We now turn to discussing the interpretation of our results and methodology and their inherent limitations that provide opportunities for future research.

The seven components are distinct in their interpretation as well as a way of estimation. They also likely differ in how likely they are reflecting the presence of any IFFs. It is not straightforward to discuss the possible presence of IFFs, or trade mispricing as understood by the existing literature, in the seven components, which partly depends on the preferred definition of IFFs. Nevertheless, we suggest that trade costs are not likely to be related to IFFs, whereas abnormal prices and the residual are likely to partly reflect some IFFs. The other components might but need not include IFFs. Across all components, it is impossible to determine with precision the share for which IFFs were responsible, if any. Still, we believe this decomposition sheds new light into the likely scale and potential channels of IFFs and, in this respect, our methodology improves on the previously researched aggregate trade reporting gaps. Similarly, with regard to trade mispricing, it is also impossible to approximate the scale with our methodology. While there is other existing evidence of transfer mispricing between related parties, there is not much evidence on trade mispricing between unrelated parties. This is important since only around one third of international trade is estimated to happen within MNEs (Shaxson, 2019). These estimates mostly rely on the US data only, but they provide us with approximate understanding of how large share of the overall global trade can be vulnerable to transfer and trade mispricing, respectively.

In interpreting our results we call for a caution since, as any researchers working on IFFs, we are faced with significant data and methodology challenges. Some of them are related to the detailed nature of the data we use. For some components (4, 5, 6 and 7, but not so for 1, 2, 3, where all the gaps have the same sign), we need to combine negative and positive values of estimated gaps at the disaggregated levels. When aggregating the detailed component results across commodities and countries, we sum them up as absolute values. This is a preferred way of aggregation and it enables us to preserve the consistency from the lower up to the highest levels of aggregation. This approach of adding absolute values of estimated gaps does not suffer from summing the potential cases of summing negative and positive gaps resulting into a zero estimated gap at higher levels of estimation (this reflects our argument that there is no such thing as net trade reporting gap). While most previous research, such as Kellenberg & Levinson (2019), usually lets any misreporting at the lower levels of aggregation to be netted out by using the aggregate at the country level only and thus label their estimates as conservative, we investigate the various reasons behind misreporting at the most disaggregated level as carefully as the available data allows us. In this way we believe we thus improve on previous approaches and provide more realistic estimates, however approximate they still are. But it also implies that our estimates at the aggregate levels are not directly comparable with previous research. Another remaining challenge, well-known in the existing literature, is that even the most detailed HS6 trade classification codes are too wide and include a range of products within one of the more than 5000 product categories. Except for using the most detailed categorisation, there is not much we can do about it, but we do acknowledge its potential confounding effects on our estimates.

There is also a number of research questions and extensions that naturally follow up from our paper. Some of them should improve the robustness of the methodology. For example, a variety of assumptions could be used in the case of trade costs or abnormal prices. We could also first adjust the prices for trade costs and tariff costs and only then for abnormal prices. Unmatched exports could be attributed not only to importers, but also to exporters and possibilities could be explored as to how the unmatched data could be decomposed in a similar way to the matched data. The role of matched trades of tiny value might be investigated. We could use the leftover transit flows from the country misclassification component to adjust the matched trade flows and capture the volumes of transit trade more precisely. In addition, the results could be extended beyond the current period of 2010-2015 (possible since 2002 and even earlier, 1998 or so, with an earlier classification). We could also study the heterogeneity across goods and identify the trading partners and products for audit purposes by customs or other authorities (e.g. which products and trading partners have disproportionately larger trade reporting gap components). Last but not least, we could analyse the country-level characteristics' determinants of the individual gap components, perhaps similarly to the more aggregate approach of Kellenberg & Levinson (2019). All these future avenues have a good research potential and policy relevance. Figure 1. Comparison of overall trade reporting gap on different levels of aggregation, 2015

Figure 2. Trade reporting gap and its components as shares overall trade gap (%) by income groups, 2015

	Country misclassification	Product misclassification	Unmatched	Abnormal prices	Trade costs	Residual
1	Greenland	Kiribati	Kuwait	Georgia	Botswana	Ethiopia
2	Cape Verde	Afghanistan	Bahamas	Lesotho	Italy	Dominican rep.
3	Central African Rep.	Lesotho	Bermuda	Botswana	USA	Palestinian terr.
4	Swaziland	Aruba	St. Lucia	Jordan	Canada	Czech Republic
5	New Caledonia	Greenland	Saudi Arabia	Cameroon	Romania	Poland
6	St. Vincent	Gambia	Zambia	Azerbaijan	Hong Kong	Mexico
7	Samoa	Australia	St. Vincent	Philippines	Portugal	Slovakia
8	Sierra Leone	Myanmar	Azerbaijan	Sierra Leone	Spain	Malaysia
9	Gambia	Solomon Island	Antigua & Barbuda	Sao Tomé & Principe	Croatia	Bangladesh
10	Afghanistan	Qatar	Aruba	Switzerland	Belgium	Brunei
11	Seychelles	Honduras	Niger	Macau	United Kingdom	Hong Kong
12	Burundi	Cambodia	Panama	Zambia	South Korea	Serbia
13	Mozambique	Yemen	Malta	Kyrgyzstan	Japan	Tanzania
14	Bahamas	Denmark	South Africa	Japan	Peru	Kyrgyzstan
15	Kiribati	Seychelles	St. Kitts & Nevis	Bermuda	Belarus	Malawi

 Table 4. Top 15 countries in each decomposition category, share of total trade gap, 2015

Source: Authors

Table 5. A tentative indicator of trading hubs based on dollar value of unmatched exports,world, top 20 countries, 2015

Country	Total reported trade	Unmatched exports (bn. USD)	Ratio of the two
	(bn. USD)		previous columns (%)
China	1483.6	13.8	0.9
Panama	10.2	11.1	109.3
South Korea	413.4	11.1	2.7
Singapore	269.1	10.5	3.9
United States	2247.1	7.4	0.3
Saudi Arabia	160.3	7.0	4.4
United Kingdom	616.9	6.6	1.1
Switzerland	245.4	5.7	2.3
Vietnam	152.4	5.3	3.5
Thailand	193.5	5.2	2.7
Japan	592.1	5.0	0.8
Bahamas	3.0	4.9	164.9
Malta	6.5	4.9	76.0
Russia	174.4	4.7	2.7
United Arab Emirates	171.8	4.6	2.7
Hong Kong	522.4	4.5	0.9
Brazil	163.1	4.5	2.8
Chile	60.4	3.9	6.5
Oman	24.3	3.5	14.6
Malaysia	163.9	3.0	1.8

Source: Authors

7 Conclusion

Trade mispricing is a real phenomenon, but the scale of these illicit financial outflows countries at all income levels is unclear. In this paper we shed more light on this by critically reviewing the existing methodologies to estimate trade mispricing and by carrying out an explorative analysis using the most extensive cross-country international trade data set. While our empirical results are not able to provide estimates of IFFs or trade mispricing, they do decompose the nominally large trade reporting gap into seven distinct components. Only further research will show how much IFFs might be responsible for each, if any, of these, but the already these results confirm the hypothesis of some of the existing literature that the potential for trade mispricing is large and visible in the data. However, since we cannot speak with our methodology about the realisation of this potential, we at least point to avenues for further research, as discussed in specific terms at the end of the results section. Still, we believe that the empirical decomposition of trade reporting gap that we develop in this paper, while somewhat orthogonal to the literature estimating IFFs and trade mispricing, can spur a new research area that might ultimately inform the question of the scale of IFFs as well.

There are at least three tentative conclusions we can preliminarily draw from our existing estimates. While these estimates would benefit from further research and they should thus be interpreted with caution, we hope even in their current form our results and the associated conclusions can usefully inform the future research. First, trade reporting gap is large, in absolute values as well as relative to the overall trade. Also, the trade reporting gap estimated at various levels of aggregation reveals that the gap substantially increases the more detailed the aggregation is. We consider the most detailed aggregation preferable and use it in our decomposition results. Second, conceptually well-defined components such as product and country misclassifications account only for a small share of trade reporting gap and most of the trade reporting gap consists of the explicitly unexplained residual component.

Third, there are substantial differences across income groups: high-income countries account for much of the trade reporting gap, while the in absolute values tiny low-income countries' gap has the highest ratio relative to their GDP. More surprisingly, the low-income countries' gap consists relatively more of unmatched trade, country misclassification and product misclassification components. Why is it so and what it implies is beyond the possibilities of this paper and warrants further research. Additionally, we observe that the empirical estimates are sensitive to the choice of assumptions and future research should provide a robustness check analysis. More generally, we consider our paper as hopefully initiating a new area of research focused on decomposition rather than on individual components only and presented empirical results should thus be considered as preliminary with follow-up research focused on validating and extending our findings.

We conclude with a brief discussion of related policy recommendations, although they are not directly related to our empirical estimates. Various global standards and actions to combat IFFs are in force or under discussion. Some relate to general policy improvements in tax, transparency and international cooperation, such as more widespread implementation of public beneficial ownership registers for companies, trusts and foundations or country-by-country reporting by multinational companies, disclosed publicly. These recommendations have been widely embraced in recent years, for example by ECA (2018). Conventionally, there are other policy initiatives more directly applied to trade mispricing. These include the transfer pricing rules prescribed by the OECD, such as the recent G20-mandated Base Erosion and Profit Shifting plan, that has been able to prolong the life of the rules further, despite calls for a more substantial reform from increasing number of individual countries and to a growing extent by the UN (FACTI, 2021).

8 References

- Baker, R. (2005). Capitalism's Achilles Heel. Dirty Money and How to Renew the Free-Market System. Wiley.
- Barbier, E. B., & Burgess, J. C. (2020). Sustainability and development after COVID-19. World Development, 135, 105082. https://doi.org/10.1016/j.worlddev.2020.105082
- Bernard, A. B., Jensen, J. B., & Schott, P. K. (2006). Transfer Pricing by U.S.-Based Multinational Firms (Working Paper No. 12493). National Bureau of Economic Research. https://doi.org/10.3386/w12493
- Boyce, J. K., & Ndikumana, L. (2001). Is Africa a net creditor? New estimates of capital flight from severely indebted sub-Saharan African countries, 1970-96. *Journal of Development Studies*, 38(2), 27–56.
- Brandt, K. (2020). Illicit financial flows and the Global South.
- Butkiewicz, J. L., & Gordon, L.-R. C. (2013). The Economic Growth Effect of Offshore Banking in Host Territories: Evidence from the Caribbean. World Development, 44, 165–179. https://doi.org/10.1016/j.worlddev.2012.09.017
- Carbonnier, G., & Zweynert de Cadena, A. (2015). Commodity Trading and Illicit Financial Flows. *International Development Policy* | *Revue internationale de politique de développement*. http://journals.openedition.org/poldev/2054
- Chalendard, C., Raballand, G., & Rakotoarisoa, A. (2017). The use of detailed statistical data in customs reforms: The case of Madagascar. *Development Policy Review*.
- Clausing, K. A. (2003). Tax-motivated transfer pricing and US intrafirm trade prices. *Journal of Public Economics*, 87(9–10), 2207–2223. https://doi.org/10.1016/S0047-2727(02)00015-4
- Cobham, A. (2005). *Tax evasion, tax avoidance and development finance* (No. 129; Queen Elizabeth House, Working Paper Series, pp. 1–20). University of Oxford, Queen Elizabeth House. http://www3.qeh.ox.ac.uk/pdf/qehwp/qehwps129.pdf

- Cobham, A. (2014). The Impacts of Illicit Financial Flows on Peace and Security in Africa. *Tana High Level Forum on Peace and Security in Africa Discussion Paper*. http://www.tanaforum.org/y-file-store/2014_Docs/IFFs%20and%20Security.pdf
- Cobham, A., & Janský, P. (2020). Estimating Illicit Financial Flows: A Critical Guide to the Data, Methodologies, and Findings. Oxford University Press. https://doi.org/10.1093/oso/9780198854418.001.0001
- Cristea, A. D., & Nguyen, D. X. (2016). Transfer Pricing by Multinational Firms: New Evidence from Foreign Firm Ownerships. *American Economic Journal: Economic Policy*, 8(3), 170–202.
- Davies, R. B., Martin, J., Parenti, M., & Toubal, F. (2017). Knocking on Tax Haven's Door: Multinational Firms and Transfer Pricing. *The Review of Economics and Statistics*. https://doi.org/10.1162/REST a 00673
- De Boyrie, M. E., Pak, S. J., & Zdanowicz, J. S. (2005). Estimating the magnitude of capital flight due to abnormal pricing in international trade: The Russia–USA case. *Accounting Forum*, *29*(3), 249–270.
- de Boyrie, M. E., Pak, S., & Zdanowicz, J. S. (2005). The Impact of Switzerland's Money Laundering Law on Capital Flows Through Abnormal Pricing in International Trade. *Applied Financial Economics*, 15(4), 217–230.
- ECA. (2018a). A study on the global governance architecture for combating illicit financial flows (pp. 1–60). Economic Commission for Africa. https://www.uneca.org/publications/study-global-governance-architecture-combating-illicit-financial-flows
- ECA. (2018b). Base Erosion And Profit Shifting In Africa: Reforms to Facilitate Improved Taxation of Multinational Enterprises (pp. 1–81). Economic Commission for Africa. https://www.uneca.org/publications/base-erosion-and-profit-shifting-africa-reforms-facilitateimproved-taxation
- Erskine, A. (2018). 3-part series on Illicit Financial Flows: Making Sense of Confusion. https://erskinomics.com/2018/10/12/3-part-series-on-illicit-financial-flows-making-sense-ofconfusion/

FACTI. (2021). FACTI Panel Report. https://www.factipanel.org/report

- Flaaen, A. (2017). The Role of Transfer Prices in Profit-Shifting by US Multinational Firms: Evidence from the 2004 Homeland Investment Act. *FEDS Working Paper*, 2017(55).
- Fontana, L. B., & Oldekop, J. A. (2020). The sustainable development goals: The bumpy road ahead. World Development, 127, 104770. https://doi.org/10.1016/j.worlddev.2019.104770
- Forstater, M. (2016). Illicit Flows and Trade Misinvoicing: Are we looking under the wrong lamppost? CMI Insight.
- Fuest, C., & Riedel, N. (2012). Tax Evasion and Tax Avoidance: The Role of International Profit
 Shifting. In P. Reuter (Ed.), *Draining Development? Controlling Flows of Illicit Funds from Developing Countries* (pp. 109–142).
 https://openknowledge.worldbank.org/handle/10986/2242
- Gara, M., Giammatteo, M., & Tosti, E. (2019). Magic mirror in my hand... How trade mirror statistics can help us detect illegal financial flows. *The World Economy*, *42*(11), 3120–3147.
- Global Financial Integrity. (2019). *Illicit Financial Flows to and from 148 Developing Countries:* 2006-2015. https://www.gfintegrity.org/report/2019-iff-update/
- Hampton, M. P., & Christensen, J. (2002). Offshore pariahs? Small island economies, tax havens, and the re-configuration of global finance. *World Development*, *30*(9), 1657–1673.
- Hogg, A., McNair, D., & Pak, S. (2009). False profits: Robbing the poor to keep the rich tax-free (Christian Aid Report). Christian Aid. http://www.christianaid.org.uk/Images/false-profits.pdf
- Hong, K. P., & Pak, S. J. (2017). Estimating Trade Misinvoicing from Bilateral Trade Statistics: The Devil is in the Details. *The International Trade Journal*, 31(1), 3–28.
- Janský, P. (2015). Updating the Rich Countries' Commitment to Development Index: How They Help Poorer Ones Through Curbing Illicit Financial Flows. *Social Indicators Research*, 124(1), 43– 65. https://doi.org/10.1007/s11205-014-0779-3
- Janský, P., Meinzer, M., & Palanský, M. (2021). Is Panama really your tax haven? Secrecy jurisdictions and the countries they harm. *Regulation & Governance*. https://doi.org/10.1111/rego.12380

- Johannesen, N., Tørsløv, T., & Wier, L. (2020). Are Less Developed Countries More Exposed to Multinational Tax Avoidance? Method and Evidence from Micro-Data. *The World Bank Economic Review*, 34(3), 790–809. https://doi.org/10.1093/wber/lhz002
- Kellenberg, D., & Levinson, A. (2019). Misreporting trade: Tariff evasion, corruption, and auditing standards. *Review of International Economics*, 27(1), 106–129. https://doi.org/10.1111/roie.12363
- Liu, L., Schmidt-Eisenlohr, T., & Guo, D. (2017). International transfer pricing and tax avoidance:
 Evidence from linked trade-tax statistics in the UK. Oxford University Centre for Business
 Taxation, Working Paper Series, 17(15).
 https://www.sbs.ox.ac.uk/sites/default/files/Business_Taxation/Docs/Publications/Working_P
 apers/Series_17/WP1715.pdf
- Murphy, R., & Christensen, J. (2005). *The Price of Offshore*. Tax Justice Network. http://www.taxjustice.net/cms/upload/pdf/Price_of_Offshore.pdf
- Nesvetailova, A., Palan, R., Petersen, H., & Phillips, R. (2021). *IFFs AND COMMODITY TRADING: OPPORTUNITIES FOR IDENTIFYING RISKS IN ENERGY TRADERS'FINANCIAL CONDUCT USING GROUPS'CORPORATE FILINGS.*
- Nicolaou-Manias, K. (2016). Trade mispricing for five African countries (Policy Briefing). GEG Africa.
- Nitsch, V. (2012). Trade mispricing and illicit flows. In Draining Development? Controlling Flows ofIllicitFundsfromDevelopingCountries.https://openknowledge.worldbank.org/handle/10986/2242
- Nitsch, V. (2016). *Trillion dollar estimate: Illicit financial flows from developing countries* (Darmstadt Discussion Papers in Economics No. 227). Darmstadt University of Technology, Department of Law and Economics. http://econpapers.repec.org/paper/zbwdarddp/227.htm
- Nitsch, V. (2017). Trade Misinvoiving in Developing Countries. CGD Policy Paper, 103. https://www.cgdev.org/sites/default/files/trade-misinvoicing-developing-countries.pdf

- Oxfam. (2000). Tax Havens: Releasing the hidden billions for poverty eradication. http://policypractice.oxfam.org.uk/publications/tax-havens-releasing-the-hidden-billions-for-povertyeradication-114611
- Picciotto, S. (2013). Is the International Tax System Fit for Purpose, Especially for Developing Countries? *International Centre for Tax and Development Working Paper*, 2013(13).
- Ponsford, V., & Mwiinga, I. (2019, January 31). Zambia hones in on fiscal sweet spot for mining. Extractive Industries Transparency Initiative. https://eiti.org/blog/zambia-hones-in-on-fiscal-sweet-spot-for-mining
- Schneider, F. (2005). Shadow Economies of 145 Countries all over the World: What do we really know? (CREMA Working Paper, Center for Research in Economics, Management and the Arts, pp. 1–58). Center for Research in Economics, Management and the Arts (CREMA). http://www.crema-research.ch/papers/2005-13.pdf
- Shaxson, N. (2019). Over a third of world trade happens inside multinational corporations. https://www.taxjustice.net/2019/04/09/over-a-third-or-more-of-world-trade-happens-insidemultinational-corporations/, https://www.taxjustice.net/2019/04/09/over-a-third-or-more-ofworld-trade-happens-inside-multinational-corporations/
- Shepherd, B. (2010). Geographical Diversification of Developing Country Exports. *World Development*, 38(9), 1217–1228. https://doi.org/10.1016/j.worlddev.2010.02.005
- Spanjers, J., & Salomon, M. (2017a). *Illicit Financial Flows to and from Developing Countries: 2005-2014* (pp. 1–51). Global Financial Integrity.
- Spanjers, J., & Salomon, M. (2017b). Illicit Financial Flows to and from Developing Countries: 2005-2014 (pp. 1–51). Global Financial Integrity.
- Tajaddini, R., & Gholipour, H. F. (2018). Control of Corruption and Luxury Goods Consumption. *Kyklos*, 71(4), 613–641. https://doi.org/10.1111/kykl.12188
- Tax Justice Network. (2007). Closing the floodgates-collecting tax to pay for development [Commissioned by the Norwegian Ministry of Foreign Affairs and Tax Justice Network]. Tax Justice Network.

- Transparency International. (2004). *Global Corruption Report 2004*. Pluto Press. http://files.transparency.org/content/download/479/1974/file/2004_GCR_PoliticalCorruption_ EN.pdf
- United Nations Economic Commission for Africa & African Union. (2015). Report of the High Level Panel on Illicit Financial Flows from Africa ("Mbeki Report"). www.uneca.org/sites/default/files/PublicationFiles/iff_main_report_26feb_en.pdf
- UNODC, & UNCTAD. (2020). Conceptual Framework for the Statistical Measurement of Illicit Financial Flows. https://www.unodc.org/documents/data-andanalysis/statistics/IFF/IFF Conceptual Framework for publication 15Oct.pdf
- Vicard, V. (2015). Profit Shifting Through Transfer Pricing: Evidence from French Firm Level Trade Data (SSRN Scholarly Paper ID 2614864). Social Science Research Network. http://papers.ssrn.com/abstract=2614864
- Wier, L. (2020). Tax-motivated transfer mispricing in South Africa: Direct evidence using transaction
 data. Journal of Public Economics, 184, 104153.
 https://doi.org/10.1016/j.jpubeco.2020.104153
- World Customs Organisation. (2018). Illicit Financial Flows via Trade Mis-invoicing Study Report 2018. http://www.wcoomd.org/-/media/wco/public/global/pdf/media/newsroom/reports/2018/wco-study-report-on-

iffs_tm.pdf?la=en

Zdanowicz, J. S. (2009). Trade-based money laundering and terrorist financing. *Review of Law and Economics*, 5(2), 854–878.

9 Appendix 1: Additional results

Table A1. Decomposing the trade reporting gap, imports, country-level results, 2015

The large table is available from the authors upon request.

Source: Authors on the basis of UN Comtrade.

Figure A1. Trade gap and its components, world, 2010 - 2015

Figure A2. Trade gap and its components as shares of GDP (%) by income groups, 2015

Source: Authors on the basis of UN Comtrade.

Figure A3. Trade gap and its components by income groups, 2015

Figure A4. Trade gap and its components by regions, 2015

10 Appendix 2: Descriptive statistics

To get a general perspective on the contents of the database, we explore the statistics for individual reporters. Table A2 shows all the reporting jurisdictions ordered by trade openness (overall trade / GDP). Except for a few small and very open economies in Central Europe and Baltics, the table offers expected pattern with the world large trading hubs in the top. Hong Kong, Singapore, Belgium, and Netherlands all occur in the top 15 most open economies. When we group the countries by geographic regions (Table A3) and levels of per capita income (Table A4) as defined by the World Bank, we get more general perspective on where the trade takes place. Europe, Central and East Asia nominally account for more than 70% or the world commodity trade, while Africa, Middle East, and South Asia together only add up to 14%. Table A4 documents that the dominant share of trade takes place in upper-middle and high-income countries. The intensity of trade relative to GDP, however, fluctuates around 0.4 in all the income country groups.

Next, we have a look at the variety of the trade by countries. We order the countries based on their shares in the world trade of individual HS 6-digit level and count their occurrences in top 20% of traders and top 5 most important traders. There are three countries which stand out in these statistics and these are, United States, China, and Germany. While these countries export or import almost 80% of the monitored categories, there is a huge gap in the occurrences thereafter. This drop is somewhat mitigated when we consider top 20% quantile of the traders instead, but the US, China, and Germany still stand out. The picture is quite similar for another check where we count occurrences of trade by a country in a category representing more than 10% of the world trade with this commodity. It turns out that all these measures correlate almost one to one with the total volume of trade in the countries (see Figure A5).

Figure A5. Number categories where the country ranks in top 5 by the share of the world trade within a category vs. overall trade volume



Source: Authors on the basis of UN Comtrade.

Country	Trade	Open.	Country	Trade	Open.	Country	Trade	Open.
Hong Kong SAR, China	1024.5	331.2	Qatar	106.9	66.1	Philippines	121.1	41.4
Singapore	604.3	198.7	Mexico	768.9	65.7	Bangladesh	77.6	39.8
Belgium	762.8	167.6	Fiji	2.8	64.2	Ecuador	39.4	39.7
Slovak Republic	141.0	161.1	Portugal	120.6	60.5	Aruba	1.1	39.5
Vietnam	310.7	160.8	Honduras	12.4	58.9	Turkey	335.0	39.0
Czech Republic	295.7	158.3	Iceland	10.0	58.9	New Zealand	69.0	38.9
Hungary	190.2	154.8	Morocco	59.1	58.4	Ethiopia	25.0	38.7
Austria	568.9	148.9	Тодо	2.4	58.3	Azerbaijan	20.5	38.6
Lithuania	53.3	128.5	Mauritius	6.8	58.3	Greece	75.1	38.4
Estonia	28.7	127.2	Montenegro	2.4	58.3	Israel	114.2	38.2
Malaysia	359.4	121.1	Bolivia	18.4	55.9	Greenland	1.0	37.9
Slovenia	52.0	120.6	Denmark	167.1	55.4	United Kingdom	1081.0	37.5
Netherlands	869.3	114.7	Sweden	273.1	54.8	Panama	20.2	37.3
Macedonia, FYR	10.8	107.5	Albania	6.2	54.7	Benin	3.1	37.3
Bulgaria	53.9	107.3	Malawi	3.4	53.1	Cyprus	7.3	37.2
Seychelles	1.4	104.9	Cote d'Ivoire	20.9	52.8	Peru	70.5	37.1
Bahrain	32.6	104.8	Canada	823.1	52.8	Dominican Republic	25.5	37.1
Namibia	12.1	104.0	Luxembourg	30.3	52.5	Sri Lanka	28.8	35.7
Cambodia	18.5	102.8	Algeria	86.3	52.0	Central African Republic	0.6	34.9
Thailand	401.2	100.0	Paraguay	18.5	51.1	China	3763.8	34.0
Belarus	55.4	98.0	St. Vincent and the Grenadines	0.4	50.0	Russian Federation	464.2	33.9
Malta	10.0	95.0	Palau	0.1	49.8	Nepal	7.3	33.9
Botswana	13.5	93.5	Chile	120.9	49.6	Indonesia	284.8	33.1
Latvia	25.2	93.5	Madagascar	4.8	49.4	Kuwait	37.7	32.9
Moldova	5.9	90.7	Maldives	2.0	49.4	Cameroon	10.1	32.6
Bosnia and Herzegovina	13.8	85.1	Burkina Faso	5.1	49.4	Saudi Arabia	209.4	32.0
Serbia	31.5	84.9	West Bank and Gaza	6.2	48.7	India	639.7	30.4
Ukraine	75.3	82.8	Myanmar	29.1	48.7	Antigua and Barbuda	0.4	30.2
Nicaragua	10.4	82.6	Sao Tome & Principe	0.2	47.9	Bahamas	3.5	29.7
Kyrgyz Republic	5.5	82.4	Spain	572.9	47.8	Colombia	86.9	29.6
Trinidad and Tobago	19.9	81.7	Lao PDR	6.8	46.9	Rwanda	2.4	29.4

 Table A2. Commodity trade openness (commodity trade as a share of GDP) and total

 commodity trade values by country. Trade values in bn. USD (2015)

Poland	381.2	79.9	Cabo Verde	0.7	46.9	Uruguay	15.6	29.2
Switzerland	542.1	79.8	Norway	181.2	46.9	Egypt, Arab Rep.	95.3	28.6
United Arab Emirates	278.9	77.9	Italy	855.8	46.7	Uganda	7.7	28.2
Tunisia	33.1	76.7	South Africa	144.1	45.4	Japan	1190.2	27.1
Mozambique	11.1	74.8	St. Lucia	0.7	45.2	Australia	342.9	25.4
Solomon Islands	0.9	74.4	Senegal	8.0	45.1	Macao SAR,	11.0	24.2
Belize	1.3	74.0	Tanzania	20.5	44.9	Pakistan	65.1	24.1
Oman	50.6	73.5	Costa Rica	24.6	44.9	Burundi	0.7	21.7
Romania	130.1	73.1	Niger	3.2	44.7	United States	3747.0	20.7
Zambia	15.4	72.7	Armenia	4.7	44.6	Brazil	356.4	19.8
Mongolia	8.5	72.0	Guatemala	28.1	44.1	Argentina	114.2	19.2
Jordan	26.9	71.6	Barbados	2.0	44.0	Yemen, Rep.	6.7	14.8
Brunei Darussalam	9.1	70.3	Sierra Leone	1.9	43.9	Sudan	14.0	14.4
Germany	2319.5	68.7	Zimbabwe	8.7	43.4	New Caledonia	3.6	-
Ireland	197.4	67.9	Jamaica	6.1	43.4	French Polynesia	1.6	-
El Salvador	15.7	67.8	France	1041.9	42.7	Bermuda	0.9	-
Korea, Rep.	934.2	67.6	Afghanistan	8.3	41.6	Iraq	-	-
Georgia	9.4	67.4	Kazakhstan	76.5	41.5	Cayman Islands	-	-
Croatia	33.1	67.0	Angola	48.1	41.4			

Source: Authors on the basis of UN Comtrade.

 Table A3. Trade openness by regions, 2015

Region	Trade (bn. USD)	GDP (bn. USD)	Openness
East Asia & Pacific	9498	21195	0.45
Europe & Central Asia	12179	19988	0.61
Latin America & Caribbean	1782	4878	0.37
Middle East & North Africa	1154	2615	0.44
North America	4571	19680	0.23
South Asia	829	2694	0.31
Sub-Saharan Africa	396	917	0.43

Income group	Trade (bn. USD)	GDP (bn. USD)	Openness
High	20644	46729	0.44
Upper-middle	7610	19849	0.38
Lower-middle	2030	5095	0.40
Low	124	294	0.42

Table A4. Trade openness by income groups, 2015

Source: Authors on the basis of UN Comtrade.

We are interested in the prices of international trade flow both on export and import side. To learn how relatively expensive the exports are on average for a country, we construct a measure where we compare the prices of the commodity categories relative to the price of the commodity weighted by traded quantities. We then average the ranking across all the commodity categories and order the countries by their average rank. We then do the same exercise for import. The main observation for the ranking is that these country standings are highly correlated (close to 0.7), i. e. the countries which typically export for higher than average world price also import for higher prices. Interestingly, there are several countries (about 10% of the sample) which occur off diagonal in Figure A6 which plots export price ranking against the import price ranking.

Additionally, about half of these tend to have relatively high export prices and low import prices (Jamaica, Sri Lanka, Netherlands, Sweden, Vietnam) while the opposite is true for the other half (UAE, Belarus, China, Kuwait, Qatar, Turkey). Although these specific examples seem diverse, we search for a factor driving these systematic differences in pricing between countries. As we mostly see rich and developed countries in the top, we plot the ranking against GDP per capita for individual countries in Figure A7. While the correlation is far from perfect (around 0.5 in absolute value), per capita income level of countries partially explains the pricing differences.

Figure A6. Country ranking on export and import prices

Source: Authors on the basis of UN Comtrade.

Figure A7. Country ranking on export and import prices and GDP per capita

IES Working Paper Series

2021

- 1. Mahir Suleymanov: Foreign Direct Investment in Emerging Markets: Evidence from Russia since the 2000s
- 2. Lenka Nechvátalová: *Multi-Horizon Equity Returns Predictability via Machine Learning*
- *3.* Milan Scasny, Matej Opatrny: *Elasticity of Marginal Utility of Consumption: The Equal-Sacrifice Approach Applied for the Czech Republic*
- *4.* Javier Garcia-Bernardo, Petr Jansky and Vojtech Misak: *Common Agricultural Policy Beneficiaries: Evidence of Inequality from a New Data Set*
- 5. Petr Jakubik, Saida Teleu: Suspension of Insurers' Dividends as a Response to the Covid-19 Crisis: Evidence from Equity Market
- 6. Boris Fisera, Menbere Workie Tiruneh, David Hojdan: *Currency Depreciations in Emerging Economies: A Blessing or a Curse for External Debt Management?*
- 7. Vojtech Molnar: *Price Level Targeting with Imperfect Rationality: A Heuristic Approach*
- 8. Alex Cobham, Tommaso Faccio, Javier Garcia-Bernardo, Petr Jansky, Jeffery Kadet, Sol Picciotto: *A Practical Proposal to End Corporate Tax Abuse: METR, a Minimum Effective Tax Rate for Multinationals*
- 9. Evžen Kočenda, Ichiro Iwasaki: *Bank Survival Around the World: A Meta-Analytic Review*
- 10. Michal Kuchta: Scenario Generation for IFRS9 Purposes using a Bayesian MS-VAR Model
- 11. Jozef Barunik, Josef Kurka: Frequency-Dependent Higher Moment Risks
- 12. Petr Pleticha: Who Benefits from Global Value Chain Participation? Does Functional Specialization Matter?
- 13. Alex Cobham, Petr Jansky, Chris Jones, Yama Temouri: An Evaluation of the Effects of the European Commission's Proposals for the Common Consolidated Corporate Tax Base
- 14. Javier Garcia-Bernardo, Petr Jansky: *Profit Shifting of Multinational Corporations Worldwide*
- 15. Tomas Domonkos, Boris Fisera and Maria Siranova: *Income Inequality as Longterm Conditioning Factor of Monetary Transmission to Bank Interest Rates in EA Countries*
- *16.* Petr Jansky, Natalia Li: *Improving the Corruption Perceptions Index: Additional Data Sources and Their Effects*
- 17. Dominika Ehrenbergerova, Josef Bajzik, Tomas Havranek: *When Does Monetary Policy Sway House Prices? A Meta-Analysis*
- 18. Daniel Kolář: Inequality in Pre-Industrial Urban Bohemia: The City of Budweis
- 19. Barbora Malinska: Forecasting Sovereign Bond Realized Volatility Using Time-Varying Coefficients Model

- 20. Jan Pintera: *Regional Convergence in the European Union: What are the Factors of Growth?*
- 21. Jiří Witzany, Martin Diviš: Interest Rate Risk of Savings Accounts
- 22. Tommaso Faccio, Sarah Godar, Petr Janský, Oliver Seabarron: *How Much Multinational Corporations Pay in Taxes and Where: Evidence from their Country-by-Country Reports*
- 23. Ondřej Schneider: Labor Migration in the European Union: The case of Central and Eastern Europe
- 24. Alex Cobham, Javier Garcia-Bernardo, Petr Janský, Miroslav Palanský: Statistical Measurement of Illicit Financial Flows in Sustainable Development Goals: Tax Avoidance by Multinational Corporations
- 25. Petr Jakubik, Saida Teleu: Impact of EU-wide insurance stress tests on equity prices and systemic risk
- *26.* Alex Cobham, Petr Janský, Jan Mareš: *Illicit Financial Flows and Trade Mispricing: Decomposing the Trade Reporting Gap*

All papers can be downloaded at: http://ies.fsv.cuni.cz



Univerzita Karlova v Praze, Fakulta sociálních věd Institut ekonomických studií [UK FSV – IES] Praha 1, Opletalova 26 E-mail : ies@fsv.cuni.cz http://ies.fsv.cuni.cz